Author’s response to reviews

**Title:** Randomized clinical trial of medial unicompartmental versus total knee arthroplasty for anteromedial tibio-femoral osteoarthritis. The study protocol

**Authors:**

Jacob Mortensen (mortensen.jacob.f@gmail.com)

Lasse Rasmussen (Lasse.enkeboelle.rasmussen@rsyd.dk)

Svend Østgaard (sveo@rn.dk)

Andreas Kappel (andreas.kappel@rn.dk)

Frank Madsen (frank.madsen@rm.dk)

Henrik Schrøder (hemsc@regionsjaelland.dk)

Anders Odgaard (ao@knee.dk)

**Version:** 2  **Date:** 06 Mar 2019

**Author’s response to reviews:**

Figures and tables have been deleted from the main text.

Figure Legends heading has been put in the relevant spot.

Figure 1 + 2 have been cited in the correct place in the text.

A response-letter has not been made, as the comments asked me not to do this at this stage.